Author's response to reviews

Title: "Parry Romberg syndrome with a wide range of ocular manifestations and some new findings (hypoplasia of the ciliary body): a case report."

Authors:

Antonio Maria Fea (antoniomfea@gmail.com)
Vittoria Aragno (vittoria.aragno@hotmail.it)
Cristina Briamonte (cristina.briamonte@gmail.com)
Mauro Franzone (maurofranzone@virgilio.it)
Davide Putignano (edivado461@gmail.com)
Federico Grignolo (federico.grignolo@unito.it)

Version: 4
Date: 10 July 2015

Author's response to reviews: see over
8th July, 2015

Manuscript ID: code used by the journal

Manuscript title: PARRY ROMBERG SYNDROME WITH A WIDE RANGE OF OCULAR MANIFESTATIONS AND SOME NEW FINDINGS (HYPOPLASIA OF THE CILIAR BODY): A CASE REPORT.

Authors: Antonio Fea et al.

Dear Dr. Papageorgiou,

Thank you for your letter. We were pleased to know that our manuscript was rated as potentially acceptable for publication in your Journal, subject to adequate revision and response to the comments raised by the reviewers.

Based on the instructions provided in your letter, we uploaded the file of the revised manuscript in type of word processor on the journal’s website.

As you notice, we have revised the manuscript by modifying it based on the comments made by the reviewers. Accordingly, we have uploaded a copy of the original manuscript marked with all the changes made during the revision process.

Appended to this letter is our point-by-point response to the comments raised by the reviewers.

- We extensively edited the written English.
- We included all titles of serological tests.
- We reported visual acuity in Snellen form.
- We compared UBM of the left and right eye. Parts of interest are labelled.
- We had the written informed consent from the patient.
- We reviewed the whole literature.
- We checked the literature you suggested.
- We corrected the spelling errors in the table.

As you notice, we agreed with all the comments raised by the reviewers. We would like to take this opportunity to thank you for allowing us to resubmit a revised copy of the manuscript.

I hope that the revised manuscript is accepted for publication in BMC Journal.

Sincerely Yours,

Antonio Fea